Author's response to reviews

Title: Case report: orthotic treatment of adult scoliosis patients with chronic back pain

Authors:

Dino F Gallo (dino@ortholutions.de)

Version: 4 Date: 25 August 2014

Author's response to reviews:

Thank you very much for reviewing the manuscript.

I tried to clarify some of the points of reviewer 2. The comment about in or out patients was not clear for me - why should a in patient not benefit from the same orthotic treatment or experience of a out patient?

You will find the changes in the manuscript.

reviewer 1:
I added the references for the first three paragraphs.
I do not think it is confusing to present the results of the questionnaire.

The improved sagittal situation of case study 1 can be clearly observed on the clinical photo.